Author’s response to reviews

Title: Intravitreal dexamethasone implant for central retinal vein occlusion without macular edema

Authors:

Eun Choi (eunyoung.choi86@gmail.com)

Hyun Kang (HGKANG08@yuhs.ac)

Sung Lee (SUNGLEE@yuhs.ac)

Min Kim (minkim76@gmail.com)

Version: 3 Date: 16 Mar 2019

Author’s response to reviews:

Editor comment> The authors addressed each criticism in a constructive manner. The changes in the manuscript represent an improvement. However, the references should be changed to BMC Ophthalmology reference style.

Reviewer 1 comment> However, the references need to be in the recommended format e.g. 1-5, 8,16,19, 24 etc.

Response to the comments> Thank you for your essential comments. We have modified the references according to the recommended format of the journal. Please see the Reference section of the revised manuscript.